Title: Epigenetic Landscapes Suggest that Genetic Risk for Intracranial Aneurysm Operates on the Endothelium

Authors:

Kerry Poppenberg (kerrypop@buffalo.edu)
Kaiyu Jiang (kaiyujia@buffalo.edu)
Michael Tso (mtso@ubns.com)
Kenneth Snyder (ksnyder@ubns.com)
Adnan Siddiqui (asiddiqui@ubns.com)
John Kolega (kolega@buffalo.edu)
James Jarvis (jamesjar@buffalo.edu)
Hui Meng (huimeng@buffalo.edu)
Vincent Tutino (vincentt@buffalo.edu)

Version: 3  Date: 17 Sep 2019

Author’s response to reviews:

Response to Editor Comments:

Editor Comments:

1. Authors’ contributions – missing authors MKT, KVS, AHS. We note that the contributions of author MKT, KVS and AHS are missing from the Authors’ contributions statement of the Declarations. The individual contributions of ALL authors to the manuscript should be specified in this section. Guidance and criteria for authorship can be found in the online journal submission guidelines.

Authors Response: We have added MKT, KV, and HAS to the Authors’ Contributions section where appropriate. Further, we have replaced “all authors” with the initials for each author.

2. Supplemental Figure 1. Supplemental Figure 1 - Appears to be corrupted. Please provide an alternate version of the image.

Authors Response: We did not see this on our end. We have re-saved the document with the embedded files as a .doc (instead of .docx). We have also provided the original TIFF files of the supplemental figures.
3. Clean manuscript. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Authors Response: We have provided a clean version of the manuscript file. The additional files were clean versions.